Author’s response to reviews

Title: The role of the General Practitioner in managing age-related hearing loss: Perspectives of General Practitioners, patients and practice staff

Authors:

Rebecca Jane Bennett (bec.bennett@earscience.org.au)

Susan Fletcher (susanlf@unimelb.edu.au)

Nicole Conway (nicole.conway@unimelb.edu.au)

Caitlin Barr (caitlin@betterhearing.org.au)

Version: 1 Date: 28 Apr 2020

Author’s response to reviews:

Dear Editorial Office,

We have made the following changes to manuscript FAMP-D-20-00173 as per your request:

-- References are numbered and in-line citations are matched back to a reference via numbering in superscript.

-- The role the funder is clearly stated in the "funding" section of the declarations. The funder only funded the study, they had no involvement in the data collection/analysis or manuscript production. Do we need to state this explicitly?

Thank you for your support,

Bec Bennett
(on behalf of all co-authors)